Author's response to reviews

Title: Quality Assessment and Anti-obesity activity of Stellaria media (Linn.) Vill.

Authors:

neeru vasudeva Dr. (neeruvasudeva@gmail.com)
surendra kumar sharma Dr. (prof.sharmask@gmail.com)
neerja rani Dr. (neerjayadav82@gmail.com)

Version: 6 Date: 21 August 2012

Author's response to reviews: see over
- Please include an Acknowledgement section at the end of the manuscript before the reference list. Please acknowledge anyone who contributed towards the study by making substantial contributions to conception, design, acquisition of data, or analysis and interpretation of data, or who was involved in drafting the manuscript or revising it critically for important intellectual content, but who does not meet the criteria for authorship. Please also include the source(s) of funding for all authors. Authors should obtain permission to acknowledge from all those mentioned in the Acknowledgements.

Reply: Added now

- We would recommend that you ask a native English speaking colleague to help you copyedit the paper. If this is not possible, you may need to use a professional language editing service. For authors who wish to have the language in their manuscript edited by a native-English speaker with scientific expertise, BioMed Central recommends Edanz (www.edanzediting.com/bmc1). BioMed Central has negotiated a 10% discount to the fee charged to BioMed Central authors by Edanz. Use of an editing service is neither a requirement nor a guarantee of acceptance for publication. For more information, see our FAQ on language editing services at http://www.biomedcentral.com/authors/authorfaq/editing.

Reply: English language Corrected now